ECH 4323L: Process Control Lab Laboratory 1

Objective: To study the dynamics of two tanks in series Procedure: The teaching assistant will explain the procedure Lab Report: The report should be brief and should have the following sections:

1. Objective: State the objective of the experiment 2. Process Model: Develop a nonlinear model to describe the dynamics of the process (use the 5 step procedure described in the lecture).

3. Data: Report the experimental data that you got.

4. Parameter Estimation: Use the experimental data to estimate the parameters of the nonlinear model.

5. Analysis:

• Linearize the nonlinear model • Calculate the dynamic response of the linearized model • Write a MATLAB program to solve the nonlinear equations directly and compute the response of the nonlinear model.

• Compare the model predictions with the experimental data. 6. Summary and Conclusions: